The Dispersal Ecology of Rhodesian Sleeping Sickness Following Its Introduction to a New Area

Nicola A. Wardrop¹*, Eric M. Fèvre², Peter M. Atkinson¹, Susan C. Welburn³

1 Geography and Environment, University of Southampton, Highfield Campus, Southampton, United Kingdom, 2 Institute of Infection and Global Health, University of Liverpool, Leahurst Campus, Neston, United Kingdom, 3 School of Biomedical Sciences, University of Edinburgh, Chancellors Building, Edinburgh, United Kingdom

Abstract

Tsetse-transmitted human and animal trypanosomiasis are constraints to both human and animal health in sub-Saharan Africa, and although these diseases have been known for over a century, there is little recent evidence demonstrating how the parasites circulate in natural hosts and ecosystems. The spread of Rhodesian sleeping sickness (caused by *Trypanosoma brucei rhodesiense*) within Uganda over the past 15 years has been linked to the movement of infected, untreated livestock (the predominant reservoir) from endemic areas. However, despite an understanding of the environmental dependencies of sleeping sickness, little research has focused on the environmental factors controlling transmission establishment or the spatially heterogeneous dispersal of disease following a new introduction. In the current study, an annually stratified case-control study of Rhodesian sleeping sickness cases from Serere District, Uganda was used to allow the temporal assessment of correlations between the spatial distribution of sleeping sickness and landscape factors. Significant relationships were detected between Rhodesian sleeping sickness and selected factors, including elevation and the proportion of land which was "seasonally flooding grassland" or "woodlands and dense savannah." Temporal trends in these relationships were detected, illustrating the dispersal of Rhodesian sleeping sickness into more 'suitable' areas over time, with diminishing dependence on the point of introduction in concurrence with an increasing dependence on environmental and landscape factors. These results provide a novel insight into the ecology of Rhodesian sleeping sickness dispersal and may contribute towards the implementation of evidence-based control measures to prevent its further spread.

Citation: Wardrop NA, Fèvre EM, Atkinson PM, Welburn SC (2013) The Dispersal Ecology of Rhodesian Sleeping Sickness Following Its Introduction to a New Area. PLoS Negl Trop Dis 7(10): e2485. doi:10.1371/journal.pntd.0002485

Editor: Philippe Solano, IRD/CIRDES, Burkina Faso

Received June 11, 2013; Accepted September 6, 2013; Published October 10, 2013

Copyright: © 2013 Wardrop et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This study was supported by the DFID Research Into Use Programme (SCW, NAW), the Medical Research Council (PMA, SCW, NAW - project G0902445; PMA, NAW – project MR/J012343/1), the Wellcome Trust (EMF - grant number 085308), the European Union's Seventh Framework Program (FP7/2007-2013) under grant agreement n° 221948 Integrated Control of Neglected Zoonoses (ICONZ) (NAW, SCW) and a European Science Foundation Senior Investigator(s) award Investigating Networks of Zoonosis Innovation INZI (SCW). The funders had no role in study design, data collection and analysis, decision to publish or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: Nicola.Wardrop@soton.ac.uk

Introduction

Since 1987, the fatal, tsetse-transmitted parasitic disease, Rhodesian sleeping sickness, has spread to eight previously unaffected districts within Uganda [1–4]. Recent research has implicated the unrestricted long-distance movement of infected, untreated livestock from endemic areas as being responsible for this spread [5,6]. It has also been suggested that tsetse movements may have contributed to the spatial spread of Rhodesian sleeping sickness in Uganda [7,8]. Several steps are required for successful parasite invasion following this type of introduction: establishment of the local population, an increase in parasite abundance and subsequent spatial dispersal [9–11]. Disease introduction, each subsequent invasion stage and, therefore, the probability of invasion success, can be influenced by spatial heterogeneity in habitat suitability, environmental conditions and species diversity.

Due to the links between disease transmission and environmental factors, any disruption to a natural ecosystem can have detrimental effects in terms of disease spread, (re)emergence or transmission intensity [12]. Historically, large sleeping sickness epidemics have resulted from ecosystem disruption (large scale cattle depopulation due to Rinderpest [13], and periods of political and civil instability [14]). In contrast, the opposite may be true in some areas: for example, Gambian sleeping sickness has disappeared from several historical foci of disease in west Africa [15]. Dramatic ecosystem changes have occurred within Uganda over the past 50 years, with the virtual elimination of wildlife from large tracts of the country and concurrent increases in human and livestock population densities (wildlife and livestock are reservoirs for *T. b. rhodesiense*). As a result, livestock are now the major reservoir of *T. b. rhodesiense* in Uganda as few other competent host species (or tsetse blood meals) remain. This concentration of circulating parasites in a narrow host range, with which human populations come into regular contact, has likely contributed to an increased risk of transmission to humans and increased vulnerability of unaffected areas to introduction of the disease.

Following parasite introduction, as well as depending on a connectable network of suitable hosts, the probability of successful invasion depends on the suitability of the surrounding landscape [16]. The probability of a successful invasion is increased by spatial dispersion away from the site of introduction, particularly when guided by environmental cues into more suitable habitats that enhance the fitness of the organism in terms of survival and reproduction [9,11]. The influence of environmental and climatic

Author Summary

Rhodesian sleeping sickness is a deadly disease, which is spread by the bite of infected tsetse flies in sub Saharan Africa. Rhodesian sleeping sickness has been spreading into new areas of Uganda over the past 15 years; this has been linked to the movement of infected, untreated livestock (which can carry the disease). Until now, there has been little focus on the impact of environmental factors (e.g. land cover) on the spatial spread of sleeping sickness after its introduction to a new area. This work makes use of Rhodesian sleeping sickness case information from December 1998 to November 2002, from an area of Uganda which was not previously affected by the disease, along with the same number of non-sleeping sickness patients, for comparison. Relationships between Rhodesian sleeping sickness and selected factors, including elevation and land cover, were detected using statistical methods. These relationships changed over time, indicating that the disease spread into more 'suitable' areas over time, determined by landscape factors. These results provide a novel insight into the environmental factors that guide the spread of Rhodesian sleeping sickness following its introduction to a new area, and can help us to develop more effective control measures to prevent further spread.

factors, including proximity to specific types of land cover (specifically areas of wetland), on the spatial distribution of tsetse vectors and Rhodesian sleeping sickness has been demonstrated [5,17–19]. Although previous research has indicated a temporal trend in the associations between Rhodesian sleeping sickness and landscape factors following the introduction of the disease to a new area [20], the significance of such factors on the probability of T. b. *rhodesiense* introduction success and its subsequent spatial dispersal has not been investigated, despite the potential practical applications of such knowledge for targeted disease control activities. A wealth of information is available regarding the impact of environmental heterogeneity on the invasion and spread of exotic plants and animals [21,22]. Despite clear overlaps with the ecology of animal and plant invasions, much of the literature relating to the spatially heterogeneous dispersal of infectious diseases is focused on population dynamics and social interactions, largely ignoring the contribution of environmental variables [10,23,24]. An understanding of the spatio-temporal interplay between parasite transmission and the natural environment would provide important information for the prevention of future spread linked to new introductions.

We used annually stratified data to examine the spatio-temporal evolution of a Rhodesian sleeping sickness outbreak within an area of eastern Uganda which was unaffected by the disease until 1998. A land cover classification was used to characterise components of the landscape which may influence the distribution of tsetse and, thus, the spread of sleeping sickness. Temporal changes in the observed relationships between sleeping sickness and selected landscape factors were studied to provide an understanding of the factors influencing the dispersal of sleeping sickness following its introduction. This research acts as an initial bridge between the invasion ecology of animals, plants and infectious diseases, emphasising the importance of multi-faceted ecosystem attributes on the probability of transmission establishment and dispersal. The analysis can provide important information to assist with the prevention of future transmission establishment and dispersal of disease following an introduction.

Materials and Methods

Ethics statement

No patient identifiable information was recorded to maintain patient confidentiality and to adhere to the International Ethical Guidelines for Biomedical Research Involving Human Subjects. The use of these data was approved by the University of Edinburgh Research Ethics Committee. Informed consent was not obtained as secondary data were used, and all data were anonymous.

Study area

The research focused on the catchment of Serere Hospital (formerly Serere Health Centre) in Serere district (formerly part of Soroti District), which is in the Eastern region of Uganda and borders Lake Kyoga (see Figure 1 for a map of Uganda highlighting the study area). The population within the study district during the 2002 national census, predominantly composed of the Iteso and Kumam ethnic groups, was approximately 104,400, with a population density of approximately 89 people per km² [25]. Primary economic activities within the study area are subsistence farming (predominantly rain-fed) and fishing (in areas in close proximity to the lake) [26]. The cattle density within the study area was approximately 105 per km² in the 2008 livestock census, although this is likely to be higher than the cattle density during the study period (1998 to 2002) due to on-going cattle restocking activities [27]. The main livestock market in the area is Brookes Corner livestock market, which runs on a weekly basis, trading an average of 90 to 120 cattle per week.

Case control data

A matched case-control study design was used with 1:1 matching. Passively detected Rhodesian sleeping sickness case records from Serere Hospital were used as cases. Each case was matched to a suitable control from the hospital inpatient,



Figure 1. Map of Uganda, highlighting the study area (Serere and Soroti districts). Note that Serere district was part of Soroti district prior to 2010.

doi:10.1371/journal.pntd.0002485.g001

outpatient, tuberculosis and maternity ward records (in that order of preference), excluding patients with a primary diagnosis of a vector-borne disease (to prevent spatial bias in the results). Matching was carried out based on age group (<1, 1–9, 10–14, 15–19, 20–49, 50–64 and \geq 65 years), gender and month of admission. The matched study design was used to ensure that the controls were selected from the same population as the cases, thus, avoiding spatial bias. The case-control data covered the time period from December 1998 to November 2002. A hand-held global positioning system (GPS) (Garmin, Olathe, KS) was used to record the location at each village's central meeting point, as determined by the village chairperson. Part of the dataset (up to June 2000) was used to identify the local livestock market as the original point of introduction [2].

Covariate data

A land cover classification was carried out using Landsat Enhanced Thematic Mapper plus (ETM+) images from 2001 in the eCognition software Version 4.2 (Definiens, Munich, Germany) [28]. The classification defined areas of "seasonally flooding grassland" and "woodland and dense savannah", which were of interest due to the habitat requirements (prefers riverine vegetation) of the predominant vector species in the study area; *Glossina fuscipes fuscipes*. More details for the land cover classification are given in the supplementary information.

The geo-referenced case-control data were visualised using ArcMap 9.1 (ESRI, Redlands, CA) along with the land cover classification. Circular buffers of 1 km (to represent the majority of the village area) and 3 km radii (to include the areas immediately surrounding the village) were created around each village centroid. This choice of buffer size was based on previous research demonstrating the significance of wetland areas up to 3 km from the homestead for the risk of Rhodesian HAT [29]. The proportions of pixels within the buffer areas that were classified as "seasonally flooding grassland" or "woodland and dense savannah" were calculated to quantify the landscape within and surrounding the case and control villages. Population density [30], elevation [31] and predicted suitability (as a percentage) for G. f. fuscipes, the predominant vector species within the study area [32], were overlaid with the case-control data in ArcMap and their values extracted for each case-control observation. The Euclidean distance (kilometres) between each village and Brookes Corner livestock market (previously implicated in the introduction of Rhodesian sleeping sickness to the study area [2]), was calculated.

Exploratory analysis

The dataset was stratified annually to allow the separate analysis of each year and illustrate temporal patterns in the observed relationships. The four years for analysis ran from December to November and, for clarity, will be referred to as the first (December 1998 to November 1999), second (December 1999 to November 2000), third (December 2000 to November 2001) and fourth (December 2001 to November 2002) years. Unadjusted odds ratios were calculated to examine the relationships between sleeping sickness status (case or control) and each independent variable for the four years using conditional logistic regression (a variation of logistic regression which takes into account a matched study design [33,34]) implemented using the *Survival* package in R statistical software [35]. The odds ratios were plotted over time to highlight temporal changes.

Multivariate analysis

Multivariate conditional logistic regression analysis was carried out for each of the four years, using forwards stepwise methodologies, beginning with the null model in each year. At each step, the variable which resulted in the largest decrease in deviance was selected. Models were compared using Chi-squared likelihood ratio tests, and variables were retained in the model if this test was significant, and the variable was significant within the model (p-value < 0.05). Any variables which lost significance in subsequent steps were removed. The stepwise addition of interaction terms between the significant variables was carried out to assess the presence of effect modification. R^2 values were used to assess the proportion of the variability in the response variable which was described by the model (note: the maximum R^2 for conditional logistic regression is 0.5). Predicted suitability for G. fuscipes was excluded from the multivariate analysis as this variable is itself a predictive model output encompassing the effects of several covariates.

Results

A total of 240 sleeping sickness cases resident within the study area were diagnosed at Serere hospital during the study period (December 1998 to November 2002). Suitable controls could not be identified for seven cases; these were excluded (see Table 1 for the number of matched and unmatched cases from each of the four years). The spatial distribution of the 233 cases within the study area varied through the study period (Figures 2 to 5). During the first year, the cases were primarily located close to the livestock market where the parasite has previously been demonstrated to have been introduced via the trade of untreated cattle, and not far from one of the main tributaries of like Kyoga. In contrast, the controls were evenly distributed across the study area. Over the subsequent years, the distribution of controls remained evenly dispersed with no directional movement, while the spatial distribution of cases moved away from the livestock market.

Univariate analysis

All four non-land cover variables (elevation, population density, predicted suitability for *Glossina fuscipes fuscipes* and distance to the

Table 1. Number of matched and unmatched cases for each year.

Time period	Number of unmatched cases	Number of matched cases
December 1998 to November 1999	2	58
December 1999 to November 2000	4	52
December 2000 to November 2001	1	44
December 2001 to November 2002	0	79
Total	7	233

doi:10.1371/journal.pntd.0002485.t001



Figure 2. Map of case control data, year 1. True colour Landsat ETM+ composite of study area, showing locations of cases and controls (as counts for each location), December 1998–November 1999, Serere hospital and Brookes Corner livestock market. Figure adapted from [20]. doi:10.1371/journal.pntd.0002485.g002

local livestock market) were significantly correlated with the occurrence of sleeping sickness; temporal trends in the odds ratios were observed for three of them (see Table 2 and Figures 6a to 6d). Increasing population density demonstrated a protective effect in each of the four years (Odds ratios (OR) = 0.95, 0.91, 0.90 and 0.93 for years one to four respectively - this was not statistically significant in the third year). A protective effect was also observed at increasing distances from the livestock market during all four years (OR = 0.69, 0.91, 0.97 and 0.96 for years one to four respectively), although this correlation lost statistical significance in the third and fourth years as the strength of the association decreased. In the first year, areas with a higher predicted suitability for G. f. fuscipes had a significantly decreased odds of sleeping sickness occurrence (OR = 0.95), but in the years two, three and four the correlation reversed (OR = 1.04, 1.09 and 1.07 respectively). A similar temporal pattern was observed for elevation, with an increased odds of sleeping sickness in higher elevation areas in the first year (OR = 1.02), and decreased odds in

the second to fourth years (OR = 0.99, 0.96 and 0.98, respectively, although this was not statistically significant in the second year).

We considered the land cover in the areas surrounding case and control villages. Significant correlations were found between sleeping sickness occurrence and the proportions of "seasonally flooding grassland" and "woodland and dense savannah" within 1 km and 3 km buffers around villages. In year one, villages with a higher proportion of "seasonally flooding grassland" in the surrounding areas had lower odds of sleeping sickness occurrence, but in years two to four the relationship reversed. This correlation had increased statistical significance using a buffer of 3 km radius (OR = 0.94, 1.01, 1.11 and 1.08 for years one to four, respectively,)although this was not statistically significant in year two). The presence of a higher proportion of "woodland and dense savannah" within areas surrounding the villages demonstrated a protective effect, with increased significance for the 1 km radius buffers (OR = 0.98, 0.93, 0.91 and 0.92 for years one to four respectively,)although this was not statistically significant for year one).



Figure 3. Map of case control data, year 2. True colour Landsat ETM+ composite of study area, showing locations of cases and controls (as counts for each location), December 1999–November 2000, Serere hospital and Brookes Corner livestock market. Figure adapted from [20]. doi:10.1371/journal.pntd.0002485.g003

Multivariate analysis

Multivariate conditional logistic regression resulted in three different final regression models (see Table 3). The model for years one and two contained distance to the livestock market (OR = 0.48, p < 0.005 for year one, OR = 0.79, p < 0.005 for yeartwo), which had a protective effect at greater distances (the effect was smaller in the second than the first year), and elevation (OR = 0.91, p = 0.02 for year one, OR = 0.95, p < 0.005 for yeartwo) which demonstrated a protective effect in areas with higher elevation. The final model for year three included the proportion of "woodland and dense savannah" within a 1 km buffer (OR = 0.88, p < 0.005) which demonstrated a protective effect in areas with a higher proportion, and the proportion of "seasonally flooding grassland" within a 3 km buffer (OR = 1.18, p = 0.01) which resulted in an increased odds of sleeping sickness in areas with a higher proportion. For the fourth year, the final model contained the proportion of "woodland and dense savannah" within a 1 km buffer (OR = 0.90, p< 0.005) which demonstrated a protective effect in areas with a higher proportion, and distance to

the livestock market (OR = 0.90, p = 0.01), which resulted in lower odds of sleeping sickness occurrence in areas further from the market (this effect was smaller in year four than years one or two). The R^2 for each model differed, with the largest R^2 for year one (0.41 compared with 0.21, 0.26 and 0.18 for years two to four, respectively).

Discussion

The temporal analysis identified the influence of selected landscape factors on the spatially heterogeneous dispersal of Rhodesian sleeping sickness following its introduction to a previously unaffected area. We analysed four annual snapshots to show the nature of the spatial dispersal of Rhodesian sleeping sickness into the landscape over time, following introduction via a cattle market. This dispersal of disease outcomes is seen as a manifestation of an underlying ecological dispersal process of the Trypanosome parasite into suitable areas conditional upon landscape suitability, cattle density and movements, human



Figure 4. Map of case control data, year 3. True colour Landsat ETM+ composite of study area, showing locations of cases and controls (as counts for each location), December 2000–November 2001, Serere hospital and Brookes Corner livestock market. Figure adapted from [20]. doi:10.1371/journal.pntd.0002485.g004

density and movements and the spatial distribution of tsetse populations. We quantified the indirect dependence of disease outcomes on non-landscape constraints (markets) and landscape constraints (land cover); and crucially, the switch from the former to the latter spatially and the rate at which this occurs. Proximity to the point of introduction initially exerted a strong spatial control, but this correlation diminished approximately two years after first measurement, in concurrence with an increasing dependence on environmental factors as the disease flowed outward to occupy the most suitable environmental niche.

In line with previous studies, increasing population density demonstrated a protective effect [17,36], illustrating the preference of the tsetse vector for areas with lower population density and, thus, less disturbance of vegetation and potential habitats or *vice versa* the avoidance of tsetse suitable habitats by humans. However, it must be noted that changes in the population size are likely to have occurred between the study period and the year for which population density data was available (2006). These population changes are not likely to have affected the results, but should be

kept in mind. Higher predicted suitability for the vector, G. fuscipes, was shown to be a risk factor in years two to four, with a temporal trend in the relationship suggesting that parasite dispersal from the point of introduction may have been into areas more suitable for vector populations and, thus, (where humans and livestock are present) for transmission of sleeping sickness. Distance to the livestock market was negatively correlated with sleeping sickness as has been noted previously [2]. A temporal trend was also evident for this correlation; the strength of the association decreased and lost statistical significance over time, illustrating the spatial dispersal away from the initial point of introduction [2]. The reinclusion of distance to the livestock market in the multivariate model for year four provides conflicting evidence from the univariate analysis which illustrated the apparent dispersion of cases away from the livestock market over the study period. This may be indicative of a secondary parasite introduction at the livestock market following the initial introduction in (or before) 1998, although it is not possible to confirm this using the data available.



Figure 5. Map of case control data, year 4. True colour Landsat ETM+ composite of study area, showing locations of cases and controls (as counts for each location), December 2001–November 2002, Serere hospital and Brookes Corner livestock market. Figure adapted from [20]. doi:10.1371/journal.pntd.0002485.g005

Temporal trends are also apparent in the relationships between sleeping sickness status and selected landscape features. The odds ratios for elevation and the proportion of land surrounding villages that was "woodland and dense savannah" decreased over time (indicating dispersal into lower lying areas with less "woodland and dense savannah") and the odds ratio for the proportion of "seasonally flooding grassland" surrounding villages increased (indicating dispersal into areas with more "seasonally flooding grassland"). The observed trends suggest that these landscape features contributed to the spatially heterogeneous dispersal away from the point of introduction; the areas into which the disease spread may represent more 'suitable' habitats for tsetse survival and reproduction and, thus, sleeping sickness transmission. The observed temporal trend in the correlation with elevation may be related to "seasonally flooding grassland", as lower lying areas are more likely to become flooded during the wet season. Further factors which may influence the spatial distribution and dispersion of Rhodesian sleeping sickness, but which have not been considered here, include proximity to water bodies, livestock

density, climatic factors (although climatic factors are not likely to vary significantly at the scale of the study area) and tsetse abundance (tsetse suitability was considered in univariate, but not multivariate analysis). The available predicted tsetse suitability was not included in multivariate modelling due to the fact that this in itself is the result of a statistical model; no other tsetse abundance data were available for the study area. Movements of humans, livestock and tsetse within the study area are also likely to play an important role in the spatial dispersal of parasites following introduction to a new area. However, data regarding these movements were not available, and, thus, it was not possible to account for this in the analysis. Further development of the analytical framework presented should attempt to incorporate these local movements to provide a broader picture of disease dispersal. Treatment of humans and cattle for trypanosomiasis may have altered the epidemiology of Rhodesian sleeping sickness during the study period, although this is not likely to have influenced the spatial spread, or the correlations between disease occurrence and landscape factors.



Figure 6. Temporal trend in odds ratios. Odds ratios (red line) over the four years for (a) population density (top left), (b) distance to livestock market (top left), (c) predicted suitability for *G. fuscipes* (centre left), (d) elevation (centre right), (e) proportion of land within 3 km that is seasonally flooding (bottom left) and (f) proportion of land within 1 km that is "woodland and dense savannah" (bottom right), with 95% confidence interval for the odds ratio (broken line) in comparison with an odds ratio of 1, which represents no significant association (blue line). Note that 2 (b) has a different scale on the y-axis from the other plots. doi:10.1371/journal.pntd.0002485.g006

This research quantifies the influence of environmental factors on the space-time dynamics of Rhodesian sleeping sickness for the first time and acts as an initial bridge between the invasion ecology of animals and plants (which has been widely studied [21,22]), and the invasion ecology of infectious diseases (which has, thus far, been largely neglected), emphasising the importance of multifaceted ecosystem attributes on the probability of disease establishment and dispersal. It is now established that Rhodesian sleeping sickness introductions have occurred via livestock trade into markets in Uganda [2,5]. The continuing movement of untreated animals through the study area and other currently unaffected areas, in conjunction with large livestock populations and low diversity of alternative hosts, increases the likelihood of further parasite introductions.

Within the wider context of sleeping sickness epidemiology across sub-Saharan Africa, Uganda is one of 36 countries affected by sleeping sickness. Despite the progressive spread of Rhodesian sleeping sickness within Uganda, the annual number of reported cases has been falling in Uganda and elsewhere. Some countries which were previously considered to be endemic for the disease

Table 2. Univariate results for the four years.

		V	N	Waxaa dhaxaa	N
variable		rear one	rear two	rear three	rear tour
Population density	OR (95% CI)	0.95 (0.91–1.00)	0.91 (0.83–0.99)	0.90 (0.79–1.02)	0.93 (0.89–0.98)
	<i>p</i> -value	0.05	0.03	0.09	0.006
Predicted suitability for G. fuscipes	OR (95% CI)	0.95 (0.91–0.99)	1.04 (1.00–1.09)	1.09 (1.03–1.15)	1.07 (1.03–1.12)
	<i>p</i> -value	0.01	0.05	<0.005	<0.005
Elevation	OR (95% CI)	1.02 (1.00–1.04)	0.99 (0.97–1.00)	0.96(0.94–0.99)	0.98 (0.97–0.99)
	<i>p</i> -value	0.03	0.15	<0.005	0.008
Distance to livestock market	OR (95% CI)	0.69 (0.57–0.85)	0.91 (0.84–0.98)	0.97 (0.91–1.04)	0.96 (0.92–1.01)
	<i>p</i> -value	<0.005	0.009	0.44	0.11
"woodland and dense savannah" (1 km) ⁺	OR (95% CI)	0.98 (0.94–1.01)	0.93 (0.88–0.98)	0.91 <i>(0.86–0.96)</i>	0.92 (0.89–0.96)
	p-value	0.16	0.008	<0.005	<0.005
"Seasonally flooding grass" (3 km) ⁺⁺	OR (95% CI)	0.94 (0.89–0.99)	1.01 (0.97–1.06)	1.11 (1.02–1.21)	1.08 (1.02–1.14)
	p-value	0.02	0.69	0.02	0.007

Unadjusted odds ratios (OR), 95% confidence intervals (CI) and p-vales where calculated using conditional logistic regression.

⁺Proportion of land within 1 km which is "woodland and dense savannah."

⁺⁺Proportion of land within 3 km which is "seasonally flooding grassland."

doi:10.1371/journal.pntd.0002485.t002

have not reported any cases in recent years, indicating possible local elimination [37,38]. Despite reductions in annual incidence and the institution of animal based interventions to reduce the reservoir of *T. b. rhodesiense* in northern Uganda (The Stamp Out Sleeping Sickness (SOS) programme which began in 2006 to prevent further spread [39]), there is still concern of further spread: since 2004, cases of Rhodesian sleeping sickness have been reported in further districts of Uganda which were previously not affected, north west of Serere district [5]. Further spread of *T. b. rhodesiense* within Uganda is possible, thus, increased understanding of the process of spatial spread of disease is required to enable effective targeting of interventions.

Year	Variable	OR (95% CI)	p - value
One	Distance to market	0.48 (0.30–0.77)	<0.005
	Elevation	0.91 (0.84–0.99)	0.02
	*R ² =0.41, AUC=0.99		
Two	Distance to market	0.79 (0.68–0.90)	< 0.005
	Elevation	0.95 (0.93–0.98)	<0.005
	*R ² =0.21, AUC=0.86		
Three	Proportion woodland (1 km)	0.88 (0.81–0.95)	< 0.005
	Proportion flooding (3 km)	1.18 (1.04–1.33)	0.01
	$*R^2 = 0.26$, AUC = 0.90		
Four	Proportion woodland (1 km)	0.90 (0.85–0.95)	< 0.005
	Distance to market	0.90 (0.83–0.98)	0.01
	$R^2 = 0.18$, AUC = 0.84		

Table 3. Final multivariate models for each of the four years.

Odds ratios (OR), 95% confidence intervals (CI) and *p*-values, R^2 values and the area under the receiver operating characteristic curve (AUC).

*Note that the maximum R^2 value in a conditional logistic regression model is 0.5.

doi:10.1371/journal.pntd.0002485.t003

spread of Rhodesian sleeping sickness and other environmentally dependent diseases, and demonstrates, for the first time, through the evolution of sleeping sickness cases, the role played by the landscape on the possibility of Trypanosome dispersal. These findings can provide foresight for intervention and planning: an expectation of the rate and extent to which suitable areas may become affected should allow suitable areas that are proximate to an introduction site to be targeted in advance of disease dispersal. For example, tsetse traps may be utilised within suitable landscapes in close proximity to livestock markets. Future research should extend this analysis to other historical outbreaks of Rhodesian sleeping sickness following introduction and to other diseases where introduction is known to occur through a focal site.

This analysis serves to enhance our current understanding of the

Supporting Information

Checklist S1 STROBE checklist.

(DOC)

Text S1 Detailed methods for the land cover classification. (DOC)

Author Contributions

Conceived and designed the experiments: NAW EMF PMA. Performed the experiments: NAW. Analyzed the data: NAW. Contributed reagents/ materials/analysis tools: EMF SCW. Wrote the paper: NAW PMA. Interpretation of results: NAW EMF SCW PMA.

References

- Okiria R (1985) The prevalence of human trypanosomiasis in Uganda, 1970 to 1983. East Afr Med J 62: 813–816.
- Fèvre EM, Coleman PG, Odiit M, Magona JW, Welburn SC, et al. (2001) The origins of a new *Trypanosoma brucei rhodesiense* sleeping sickness outbreak in eastern Uganda. Lancet 358: 625–628.
- Batchelor N, Atkinson PM, Gething PW, Picozzi K, Fèvre EM, et al. (2009) Spatial predictions of Rhodesian Human African Trypanosomiasis prevalence in Kaberamaido and Dokolo, two newly affected districts of Uganda. PLoS Negl Trop Dis 3: e563.

- Picozzi K, Fèvre EM, Odiit M, Carrington M, Eisler MC, et al. (2005) Sleeping sickness in Uganda: a thin line between two fatal diseases. Br Med J 331: 1238– 1241.
- Wardrop N, Atkinson PM, Gething PW, Fèvre EM, Picozzi K, et al. (2010) Bayesian Geostatistical Analysis and Prediction of Rhodesian Human African Trypanosomiasis. PLoS Negl Trop Dis 4: e914.
- Fèvre EM, Picozzi K, Fyfe J, Waiswa C, Odiit M, et al. (2005) A burgeoning epidemic of sleeping sickness in Uganda. Lancet 366: 745–747.
- Aksoy S, Caccone A, Galvani A, Okedi LA (2013) *Glassina fuscipes* populations provide insights for human African trypanosomiasis transmission in Uganda. Trends Parasitol 29: 394–406.
- Beadell JS, Hyseni C, Abila PP, Azabo R, Enyaru JCK, et al. (2010) Phylogeography and Population Structure of *Glossina fuscipes fuscipes* in Uganda: Implications for Control of Tsetse. PLoS Negl Trop Dis 4: e636.
- Schreiber SJ, Lloyd-Smith JO (2009) Invasion dynamics in spatially heterogeneous environments. Am Nat 174: 490–505. DOI 10.1086/605405.
- Anderson RM, May RM (1986) The Invasion, Persistence and Spread of Infectious-Diseases Within Animal and Plant-Communities. Philosophical Transactions of the Royal Society of London Series B-Biological Sciences 314: 533–570.
- Lambin EF, Tran A, Vanwambeke SO, Linard C, Soti V (2010) Pathogenic landscapes: Interactions between land, people, disease vectors, and their animal hosts. International Journal of Health Geographics 9. 54; DOI 10.1186/1476-072X-9-54.
- Patz JA, Daszak P, Tabor GM, Aguirre AA, Pearl M, et al. (2004) Unhealthy landscapes: Policy recommendations on land use change and infectious disease emergence. Environ Health Perspect 112: 1092–1098.
- Fèvre EM, Coleman PG, Welburn SC, Maudlin I (2004) Reanalyzing the 1900– 1920 sleeping sickness epidemic in Uganda. Emerg Infect Dis 10: 567–573.
- Berrang-Ford L (2007) Civil conflict and sleeping sickness in Africa in general and Uganda in particular. Conflict and Health 1: 6.
- Courtin F, Jamonneau V, Duvallet G, Garcia A, Coulibaly B, et al. (2008) Sleeping sickness in West Africa (1906–2006): changes in spatial repartition and lessons from the past. Tropical Medicine & International Health 13: 334–344. DOI 10.1111/j.1365-3156.2008.02007.x.
- With KA (2002) The landscape ecology of invasive spread. Conserv Biol 16: 1192–1203.
- Odiit M, Bessell PR, Fèvre EM, Robinson T, Kinoti J, et al. (2006) Using remote sensing and geographic information systems to identify villages at high risk for rhodesiense sleeping sickness in Uganda. Trans R Soc Trop Med Hyg 100: 354– 362.
- Hendrickx G, Napala A, Slingenbergh JHW, De Deken R, Rogers DJ (2001) A contribution towards simplifying area-wide tsetse surveys using medium resolution meteorological satellite data. Bull Entomol Res 91: 333–346.
- Rogers DJ, Hay SI, Packer MJ (1996) Predicting the distribution of tsetse flies in West Africa using temporal Fourier processed meteorological satellite data. Ann Trop Med Parasitol 90: 225–241.
- 20. Wardrop N, Fèvre EM, Atkinson PM, Welburn S (2012) An exploratory GISbased method to identify and characterise landscapes with an elevated

epidemiological risk of Rhodesian human African trypanosomiasis. BMC Infect Dis 12: 316.

- Elton, C. S. (1958) The Ecology of Invasions by Animals and Plants. Norwich: Fletcher and Son Ltd.
- Bradley BA, Mustard JF (2006) Characterizing the landscape dynamics of an invasive plant and risk of invasion using remote sensing. Ecol Appl 16: 1132– 1147.
- Real LA, Biek R (2007) Spatial dynamics and genetics of infectious diseases on heterogeneous landscapes. Journal of the Royal Society Interface 4: 935–948. DOI 10.1098/rsif.2007.1041.
- May RM, Gupta S, Mclean AR (2001) Infectious disease dynamics: what characterizes a successful invader? Philosophical Transactions of the Royal Society of London Series B-Biological Sciences 356: 901–910.
- 25. Uganda Bureau of Statistics (2002) 2002 Uganda Population and Housing Census.
- 26. Fountain Publishers (2007) Uganda districts information handbook. Kampala: Fountain Publishers.
- 27. Uganda Bureau of Statistics (2009) National Livestock Census 2008.
- Baatz M, Benz U, Dehghani S, Heynen M, Höltje A, et al (2000) eCognition professional user guide 4. Munich: Definiens-Imaging.
- Zoller T, Fèvre EM, Welburn S, Odiit M, Coleman P (2008) Analysis of risk factors for *T. brucei rhodesiense* sleeping sickness within villages in south-cast Uganda. BMC Infect Dis 8: 88. 10.1186/1471-2334-8-88.
- 30. Oak Ridge National Laboratory (2006) Landscan 2006 Global Population Dataset.
- 31. US Geological Survey (2006) "Finished" 3-arc second SRTM Format Documentation.
- Wint W, Rogers DJ (2000) Predicted distributions of tsetse in Africa. Rome: Food and Agricultural Organization.
- Woodward M (2004) Case-control studies. In: Epidemiology Study Design and Data Analysis. Boca Raton: CRC Press. pp. 273–334.
- Breslow NE, Day NE (1980) Conditional logistic regression for matched sets. In: Davis W, editors. Statistical Methods in Cancer Research Volume 1: The Analysis of Case-Control Studies. Lyon, France: IARC Scientific Publications. pp. 247–279.
- R Development Core Team (2006) R: A language and environment for statistical computing, version Vienna, Austria: R Foundation for Statistical Computing.
- Reid RS, Kruska RL, Deichmann U, Thornton PK, Leak SGA (2000) Human population growth and the extinction of the tsetse fly. Agriculture Ecosystems & Environment 77: 227–236.
- Simarro PP, Cecchi G, Paone M, Franco JR, Diarra A, et al. (2010) The Atlas of human African trypanosomiasis: a contribution to global mapping of neglected tropical diseases. International Journal of Health Geographics 9: 57; DOI 10.1186/1476-072X-9-57.
- Simarro PP, Cecchi G, Franco JR, Paone M, Diarra A, et al. (2012) Estimating and Mapping the Population at Risk of Sleeping Sickness. PLoS Negl Trop Dis 6: e1859.
- Kabasa JD (2007) Public-private partnership works to stamp out sleeping sickness in Uganda. Trends Parasitol 23: 191–192.